Author's response to reviews

Title: A possible role for miRNA silencing in disease phenotype variation in Swedish transthyretin V30M carriers

Authors:

Malin Olsson (malin.olsson@medbio.umu.se)
Nina Norgren (nina.norgren@medicin.umu.se)
Konen Obayashi (konen@fc.kuh.kumamoto-u.ac.jp)
Violaine Plante-Bordeneuve (vplante@free.fr)
Ole B Suhr (ole.suhr@medicin.umu.se)
Kristina Cederquist (kristina.cederquist@vll.se)
Jenni Jonasson (jenni.jonasson@vll.se)

Version: 4 Date: 9 September 2010

Author's response to reviews: see over
Dear Sir or Madam

Thank you for accepting our manuscript for publication in BMC Medical Genetics.

Please find below a point-to-point description of the changes made.

Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
- I have inserted a sentence to clarify that.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above

Abstract: remove keywords. – Has been made.

Coloured text: please remove. – Has been made.

Track changes: please remove. – Has been made.

References: please provide the full citation details for reference 11.
- Have inserted the full citation details for reference 11.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
- Small changes have been made.

Sincerely

Malin Olsson PhD, Corresponding author

Medical and Clinical Genetics
Building 6M, 3rd floor
Umeå University Hospital
SE 901 85 Umeå
Sweden
E-mail: malin.olsson@medbio.umu.se
Telephone +46 90 7851767
FAX +46 90 130760